Author’s response to reviews

Title: CTHRC1 Induces Non-small Cell Lung Cancer (NSCLC) Invasion through Upregulating MMP-7/MMP-9

Authors:

Weiling He (williamhe@qq.com)
Hui Zhang (sysuzhanghui@163.com)
Yuefeng Wang (fyf1203@163.com)
Yanbin Zhou (sumszyb@163.com)
Yifeng Luo (luoyifeng@mail.sysu.edu.cn)
Yongmei Cui (cuiyongmei1989@126.com)
Neng Jiang (misterjiang123@126.com)
Wenting Jiang (1150743751@qq.com)
Han Wang (wanghan9836@163.com)
Di Xu (xudimail@126.com)
Shuhua Li (lishu2007@126.com)
Zhruo Wang (feilengchui@163.com)
Yangshan Chen (chyshan2@mail2.sysu.edu.cn)
Yu Sun (suny77@mail2.sysu.edu.cn)
Yang Zhang (zybf1208@gmail.com)
Hsian-Rong Tseng (hrtseng@mednet.ucla.edu)
Xuenong Zou (zxnong@hotmail.com)
Liangtang Wang (wanglt@126.com)
Zunfu Ke (kezunfu@mail.sysu.edu.cn)

Version: 1 Date: 02 Jul 2017
Author’s response to reviews:

Dear Dafne Solera:

On behalf of my co-authors, I am resubmitting this revised manuscript entitled “CTHRC1 Induces Non-small Cell Lung Cancer (NSCLC) Invasion and Metastasis through Upregulating MMP-7/MMP-9 (#BCAN-D-17-00259)”. We are grateful that you were willing to take your time to provide valuable comments that dramatically improve the scientific quality of our manuscript.